Author’s response to reviews

Title: Tobacco smoking and semen quality in infertile males: A systematic review and meta-analysis

Authors:
Pravesh Kumar Bundhun (pravesh021@gmail.com)
Girish Janoo (girishjanoo@hotmail.com)
Akash Bhurtu (akashbhurtu@gmail.com)
Abhishek Teeluck (abhishek_teeluck78@yahoo.com)
Mohammad Zafooruddin Soogund (zaf9@ymail.com)
Manish Pursun (manish_hardrocker@yahoo.com)
Feng Huang (huangfeng7925@163.com)

Version: 2 Date: 19 Oct 2018

Author’s response to reviews:

POINT TO POINT RESPONSE TO EDITORS AND REVIEWERS

1. I'm afraid the quality of the English used throughout your manuscript does not currently meet our requirements. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service.

AUTHOR RESPONSE: THANK YOU. THE MANUSCRIPT WAS RE CHECKED FOR LANGUAGE ISSUES. HOPE THE EDITOR IS NOW SATISFIED WITH THE LANGUAGE. THANK YOU.

2. Please remove the ‘Trial Registration’ section from page 3 of your Manuscript.

AUTHOR RESPONSE: THE SUGGESTED CHANGES HAVE BEEN MADE. THANK YOU.

3. Igor Toskin (Reviewer 3): In general my conclusion that the manuscript addresses very important public health issue (risk factors, smoking, associated with the semen quality). The manuscript is well-written. The comments provide by both reviewers have been adequately addressed. The manuscript needs some language corrections.
AUTHOR RESPONSE: THANK YOU. APPROPRIATE CORRECTIONS HAVE BEEN MADE THROUGHOUT.